Author’s response to reviews

Title: A global survey on occupational health services in selected International Commission on Occupational Health (ICOH) member countries

Authors:
Jorma Rantanen (jorma.h.rantanen@gmail.com)
Suvi Lehtinen (suvi.a.lehtinen@outlook.com)
Antonio Valenti (a.valenti@inail.it)
Sergio Iavicoli (s.iavicoli@inail.it)

Version: 1 Date: 30 May 2017

Author’s response to reviews:
We revisited the manuscript. By resubmitting our manuscript we confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.